Author's response to reviews

Title: Obesity and overweight: prevalence and associated socio-demographic factors among mothers in three different areas in the Gaza Strip-Palestine: a cross-sectional study

Authors:

Rima Rafiq El Kishawi (rimaa2_us@yahoo.com)
Wan Abdul Manan Wan Muda (wanmanan@usm.my)
Soo Kah Leng (sookl@usm.my)
Yehia Awad Abed (Yabed333@yahoo.com)

Version: 3 Date: 2 April 2014

Author's response to reviews:

Dear the Editor,

We would like to submit the enclosed manuscript “Obesity and overweight: prevalence and associated socio demographic factors among mothers in three different areas in the Gaza Strip-Palestine: a cross sectional study” for your consideration for publication in BMC Obesity Journal. The manuscript has been approved by all authors and has never been published, or under the consideration for publication elsewhere.

*Please refer to the attached files for our responses to the reviewers’ comments.

Yours Sincerely,

Rima R.R El Kishawi
PhD candidate, Nutrition Department
School of Health Sciences, Health Campus,
Universiti Sains Malaysia, Kubang Kerian,
16150 Kelantan, Malaysia
Email: rimaa2_us@yahoo.com

2nd April, 2014